Author's response to reviews

Title: U.S. healthcare providers' knowledge, attitudes, beliefs, and perceptions concerning Chronic Fatigue Syndrome

Authors:

Dana J Brimmer (dyv4@cdc.gov)
Frederick Fridinger (fwf0@cdc.gov)
Jin-Mann S Lin (dwe3@cdc.gov)
William C Reeves (wcr1@cdc.gov)

Version: 3 Date: 15 April 2010

Author's response to reviews: see over
April 12, 2010

The BMC Series Editorial Production Team
BMC-series journals
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Dear The BMC Series Editorial Production Team,

Thank you to the editors and reviewers for providing a final review of our manuscript “U.S. Healthcare Providers’ Knowledge, Attitudes, and Beliefs Concerning Chronic Fatigue Syndrome.” We appreciate the time that was required on behalf of the third reviewer.

We have made all the edits suggested by the third reviewer. We have also made all the major and minor revisions as requested by the Editorial Production Team.

Thank you